Author's response to reviews

Title: Measurement of the total antioxidant response using a novel automated method in subjects with nonalcoholic steatohepatitis

Authors:

Mehmet Horoz (mehmethoroz@yahoo.com)
Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Fusun F Bolukbas (fusunbol@yahoo.ca)
Tevfik Sabuncu (tsabuncu@harran.edu.tr)
Mehmet Aslan (m.aslan301@mynet.com)
Serpil Sarifakiogullari (serpilsarifaki@yahoo.com)
Necla Gunaydin (g.necla@mynet.com)
Ozcan Erel (erelozcan@harran.edu.tr)

Version: 9 Date: 8 October 2005

Author's response to reviews:

Dear Editor, October 8, 2005

Thanks for your attention that you have paid for our manuscript entitled as "Measurement of the total antioxidant response using a novel automated method in subjects with nonalcoholic steatohepatitis (MS: 3288188686365540)". We provided copy-editing for the manuscript. Please find attached our revised manuscript.

Sincerely Yours'
Mehmet Horoz, MD.